Author's response to reviews

Title: Elastic Stable Intramedullary Nailing (ESIN), Orthoss(R) and Gravitational Platelet Separation- System (GPS(R)): An effective method of treatment for pathologic fractures of bone cysts in children

Authors:

Marion Rapp (rapp@paedia.ukl.mu-luebeck.de)
Daniel Svoboda (daniel.svoboda@umm.de)
Lucas M Wessel (lucas.wessel@umm.de)
Martin M Kaiser (kaiser@medinf.mu-luebeck.de)

Version: 5 Date: 2 February 2011

Author's response to reviews:

Dear Editor,

Minor revisions:
- Highlighting remove: done
- Figure citation: changed

Major revision:
Individual legends for each figure: done

Sincerely yours

Marion Rapp